Author’s response to reviews

Title: Effect of nipple shield use on milk removal in mothers with pain: a mechanistic study.

Authors:

Viviane Silva Coentro (viviane.silvacoentro@research.uwa.edu.au)
Sharon L Perrella (sharon.perrella@uwa.edu.au)
Ching Tat Lai (ching-tat.lai@uwa.edu.au)
Alethea Rea (alethea.rea@murdoch.edu.au)
Kevin Murray (kevin.murray@uwa.edu.au)
Donna Tracy Geddes (donna.geddes@uwa.edu.au)

Version: 2 Date: 12 Jan 2020

Author’s response to reviews:

The University Western Australia
M310, 35 Stirling Hwy, Crawley WA,
AUSTRALIA 6009

Tovah Honor Aronin, PhD
Editor-in-Chief
BMC Pregnancy and Childbirth

January 13th, 2020

Dear Dr. Aronin,

Thank you for providing the opportunity to revise our manuscript “Effect of nipple shield use on milk removal: a mechanistic study” so it may be suitable for publication in BMC Pregnancy and Childbirth. Please see below a point-by-point response to the feedback received.

1. Please be sure to include the email address for the corresponding author on the title page. The corresponding author’s email address has been added to the title page, page 1 line 29.

2. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.
A single clean copy of the manuscript has been uploaded.

Thank you for your consideration.

Sincerely,

Sharon Perrella PhD, RN RM IBCLC